Author's response to reviews

Title: Diagnostic utility of zinc protoporphyrin to detect iron deficiency in Kenyan pregnant women

Authors:

Martin N. Mwangi (mart.mwangi@gmail.com)
Sumi Maskey (sumi.maskey@gmail.com)
Pauline E.A. Andang'o (paulango@hotmail.com)
Noel K. Shinali (kshinali@yahoo.com)
Johanna M. Roth (j.roth@kit.nl)
Laura Trijsburg (laura.trijsburg@wur.nl)
Alice M. Mwangi (mwangi.alice4@gmail.com)
Han Zuilhof (han.zuilhof@wur.nl)
Barend van Lagen (barend.vanlagen@wur.nl)
Huub F.J. Savelkoul (huub.savelkoul@wur.nl)
Ayse Y. Demir (ay.demir@meandermc.nl)
Hans Verhoef (hans.verhoef@wur.nl)

Version: 3 Date: 17 October 2014

Author's response to reviews: see over
Referee 1:
The authors in their revised manuscript give prevalence estimates for vitamin B12 and folate deficiency. These estimates are very low prevalence and below 1%. This finding deserves comment in the discussion.

This has now been discussed (L. 429-32).

Please can you also adhere to the following editorial requests as below:

(1) Was informed consent obtained? If yes please can you add "informed" to the consent statement.
This has now been done (L. 146).

(2) Can you add the date of the trial registration to the end of the abstract.
This has now been done (L. 86).

Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals). It is important that your files are correctly formatted.
We believe that the present version of the manuscript conforms to the journal style.